ENTRY RESTRICTIONS, INDUSTRY EVOLUTION AND DYNAMIC EFFICIENCY: EVIDENCE FROM COMMERICAL BANKING

Jith Jayaratne and Philip E. Strahan

Federal Reserve Bank of New York Research Paper No. 9630

August 1996

This paper is being circulated for purposes of discussion and comment only.

The contents should be regarded as preliminary and not for citation or quotation without permission of the author. The views expressed are those of the author and do not necessarily reflect those of the Federal Reserve Bank of New York or the Federal Reserve System.

Single copies are available on request to:

Public Information Department Federal Reserve Bank of New York New York, NY 10045

Entry Restrictions, Industry Evolution and Dynamic Efficiency: Evidence from Commercial Banking*

Jith Jayaratne
Federal Reserve Bank of New York
e-mail:jith.jayaratne@frbny.sprint.com

Philip E. Strahan
Federal Reserve Bank of New York
e-mail:philip.strahan@frbny.sprint.com

August 1996

Abstract

This paper shows that bank performance improves significantly after restrictions on bank expansion are lifted. We find that profits increase and loan quality improves after states permit statewide branching, and--to a lesser extent--after states allow interstate banking. The improvements following branching deregulation appear to occur because better banks increase market share at the expense of their less efficient rivals. By retarding the "natural" evolution of the industry, branching restrictions reduced the performance of the average banking asset. We also find limited support for the hypothesis that more competitive banking markets following deregulation better discipline bank managers, thereby improving bank performance.

JEL Classification: G2, L5

^{*} We thank Rebecca Demsetz, Beverly Hirtle, Jose Lopez and Marc Saidenberg for helpful comments. The opinions expressed in this paper reflect the authors' views and do not necessarily reflect the views of the Federal Reserve Bank of New York or the Federal Reserve System.

I. Introduction

How do price and entry regulations affect market structure, industry evolution, management quality, and through these, dynamic efficiency? Relatively little is known about this question. Data limitations have forced extant empirical studies of the impact of price and entry regulations to focus on changes in prices, profits and wages, and generally neglect dynamic issues of industry evolution and efficiency (Winston 1993).

In this paper, we look to the banking industry for evidence on the dynamic effects of entry regulation. The unique history of the regulation of commercial banking in the United States offers several advantages. First, banks were subjected to extremely severe entry barriers in the form of branching restrictions at a relatively early stage of the industry's development. Banks have traditionally been prevented from crossing state lines and, until the 1980s, they were unable to cross *county* lines in many states. As a result, the U.S. banking industry is extremely fragmented, with thousands of banks and bank holding companies. This structure contrasts sharply with other countries, where a few very large institutions dominate.

Second, geographic restrictions on banking have been gradually lifted over the past two decades. This allows us to use changes in banking once these restrictions are lifted to understand the effects of such restrictions. Moreover, because most branching restrictions were imposed by state regulators, and because states deregulated at different times, we are able to use states which did not change their policy regimes to control for potentially confounding effects such as the business cycle. Pooling time series and cross sectional data also allows us to control for the effects of unmeasured differences across states.

Existing literature on the effects of bank branching restrictions suggests that these regulations limited banks' ability to diversify portfolio risks (Calomiris 1993, Demsetz and Strahan 1995) and increased market power (Rhoades 1982, Flannery 1984, Evanoff and Fortier 1988, Amel and Liang 1992).

Less is known, however, about the effects of branching restrictions on bank efficiency.

The only hypothesis on this topic with empirical support is that industry efficiency may have been impaired by geographic restrictions because they vitiated corporate control markets by reducing the number of potential acquirers, thereby worsening agency problems between bank owners and managers. This may have contributed to increased costs and reduced profitability.

Schranz (1993) finds that banks in states with strict restrictions on geographic expansion are less profitable than banks in states without such restrictions. Hubbard and Palia (1995) find that managers are paid more but their compensation is tied more closely to banks' performance in states that allow interstate banking. Hubbard and Palia interpret this as evidence of a more active takeover market and better disciplined management in states with interstate banking. However, they offer no evidence that banks improved their performance following interstate banking.

In this paper, we find that banks' performance -- profitability and loan quality -improves sharply once restrictions on intrastate branching are lifted and, to a lesser extent, after
interstate banking is permitted. We show that the observed performance improvement is not
entirely the result of the timing of deregulation. If states deregulate during economic downturns,
then following deregulation their banking systems may improve simply as a consequence of
economic recovery. We account for this potentially spurious "correlation" between bank
performance and branching deregulation by controlling for states' business cycles in our

regressions. Moreover, bank performance improves significantly after branching deregulation but not before, suggesting that causality flows from deregulation to improved performance rather than the reverse. (However, bank performance improves in the period immediately before interstate banking is allowed.)

We find some evidence that the improved bank performance following deregulation is due to better disciplined management. Takeover rates increase significantly following interstate banking, consistent with the disciplining hypothesis. Moreover, lending to bank insiders, a significant indicator of management consumption activity, decreases substantially following branching deregulation.

We find stronger evidence that branching deregulation triggered a process of selection, whereby better performing banks expand at the expense of high-cost, low-profit banks. Although better-performing banks grow faster than under-achievers before intrastate branching is allowed, we find that low-cost, high-profit banks grow even faster once branching restrictions are lifted. This suggests that branching restrictions imposed binding constraints on the ability of better-managed banks to grow. Once these restrictions were lifted, better banks expanded at the expense their poorly-managed rivals, thereby improving efficiency and profitability of the average bank asset.

On balance, the changes in banking after deregulation of geographic restrictions suggest that these restrictions allowed inefficient banks to survive and worsened agency problems between owners and management, thereby lowering the quality of the average bank and of banking intermediation in the economy. The extensive literature on the importance of bank intermediation implies that the real economy may have suffered as a result. This conjecture is

confirmed in Jayaratne and Strahan (1996), who find that states' growth rates decreased as a result of branching restrictions.

The rest of the paper is organized as follows. Section II briefly summarizes the legislative history of intrastate branching and interstate banking. Section III describes the empirical methods and presents our estimates of improved banking performance. Section IV provides evidence supporting both the disciplining and selection hypotheses. Section V concludes.

II. Recent history of interstate banking and intrastate branching

Interstate banking was effectively prohibited by the Bank Holding Company Act of 1956. The Douglas Amendment to that Act prohibited bank holding companies from establishing or purchasing bank subsidiaries across state lines unless the target bank's state authorized it. Since no state allowed such transactions at the time, the 1956 Act prevented interstate banking. In 1978, Maine permitted out-of-state bank holding companies (BHCs) to buy Maine banks. By 1992, the end of the sample period used in this paper, all states but Hawaii had entered an interstate banking agreement with other states.

Many states allowing interstate banking also limited the extent of entry by out-of-state BHCs in several ways. For example, fifteen states and Washington, D.C. allowed entry only by BHCs headquartered in selected states. The restrictions attached to interstate banking may have limited the extent of interstate activity, but they have not prevented a sharp increase in such activity. The percentage of deposits held by subsidiaries of out-of-state BHCs in the typical state expanded from 2 to 28 percent between 1979 and 1994 (Berger, Kashyap and Scalise, 1995). Moreover, banks appear to demand more from management following interstate banking:

Hubbard and Palia (1995) find a stronger pay-performance relation for CEOs and greater CEO turnover rates in states with interstate banking.

In addition to prohibiting interstate banking, most states entered the 1970s with restrictions on bank branching within state borders. For example, Florida prohibited branch banking entirely until 1977, when banks were allowed to branch within the county where their main offices were located. In 1988, Florida permitted branching statewide. Only 13 states allowed unrestricted intrastate branching in 1974. During the next two and a half decades, 35 states and Washington, D.C. substantially eliminated restrictions on intrastate branching. By 1992, all but three states allowed some form of statewide branching.

Many states had allowed banking companies to expand within the state by forming multibank holding companies (MBHCs) long before they allowed branch banking. If the MBHC structure allowed banks to grow optimally without branching, we would expect branching to have had little impact. MBHCs are more costly to operate than branch banks, however, because they require multiple boards of directors and separate capitalization of each bank subsidiary. The high cost of the MBHC structure is confirmed by the fact that many multibank holding companies converted their bank subsidiaries into branches once branching was allowed (McLaughlin 1994). Other research also indicates that branch banking had important effects on the structure of banking markets. Amel and Liang (1992) find significant entry into local markets after intrastate branching restrictions are lifted via de novo branching. Calem (1994) finds that small banking companies lose market share after branching reform.

¹ Moore (1996) questions whether the decline in small banking companies is the result of branching deregulation. He shows that the rate of decline in small banking company market share does not accelerate after deregulation.

Table 1 describes the history of the deregulation of restrictions on intrastate branching and interstate banking since 1970.² The first column presents the year in which each state permitted branching via merger and acquisition (M&A) through the holding company structure. The second column presents the date at which each state first permitted interstate banking. The dates chosen in Table 1 reflect the time at which each state <u>finished</u> the branching deregulation process.³ These choices in some cases require judgment, since some states deregulated gradually over time. In four cases we chose dates earlier than the literal end of the process of deregulation since we felt that the remaining restrictions no longer imposed a meaningful constraint on branching.⁴

In addition to the two types of deregulation listed in Table 1, two more sets of geographic restrictions were also lifted by states after 1970. The intrastate branching deregulation dates listed in Table 1 are the dates when states allowed banks to expand statewide by acquiring another bank's branch or by acquiring a whole bank and converting it into branches (M&A branching). However, de novo branching — where a new branch is established — was often

² We exclude Delaware and South Dakota from our analysis because these two states created incentives for credit card banks to locate there.

³ Dates for deregulation of both branching restrictions and restrictions on interstate banking are taken from Amel (1993).

⁴ For instance, in 1982 Pennsylvania began permitting banks to branch in the home office county, in a contiguous county, in a bicontiguous county or in the counties of Allegheny, Delaware, Montgomery and Philadelphia. In 1990, Pennsylvania permitted unrestricted branching statewide. In the results presented below, we assume that by 1982 Pennsylvania permitted intrastate branching, despite the fact that the process was not finished until eight years later, since the effect of the 1982 law brought Pennsylvania so close to complete intrastate branch freedom. We follow a similar practice for the states of Ohio, Virginia and Washington. Our results are not sensitive to the alternative dating of deregulation in these four states.

prohibited even after M&A branching was allowed. Typically, states allowed de novo branching a few years after they allowed M&A branching. In this paper, we focus only on M&A deregulation because de novo branching occurred too soon to separate out its effects from those of M&A branching.

Another type of geographic restriction not examined here are the restrictions placed on the BHC expansion within states. Illinois, for example, prohibited multi-bank holding companies in 1957. However, most states did allow intrastate MBHC expansion. By 1975, the beginning of our sample period, 35 states allowed MBHC expansion within each state. Of the fifteen remaining states, all but Rhode Island relaxed MBHC restrictions between 1975 and 1992. We do not examine intrastate MBHC deregulations partly because there were so few of them, and partly because most of these states relaxed MBHC restrictions at about the same time as they relaxed branching restrictions, making it difficult to separate the effects of MBHC deregulation.

III. Deregulation and bank performance

As explained at the outset, we are interested in using changes in profits and loan quality following deregulation to infer what effect restrictions on geographical expansion had on the efficiency of the banking system. Limits to bank expansion could have reduced efficiency in the industry for at least two reasons. First, in an environment in which banks cannot easily branch into new markets, the growth opportunities available to the better managed banks are artificially constrained. After these constraints are lifted, better banks can expand at the expense of poorly-managed banks, thereby improving the overall efficiency of the banking system. According to this selection hypothesis, the average quality of banking will improve after deregulation because the better-run, more efficient banks gain market share relative to their less efficient competitors.

Second, limits to branching and mergers and acquisitions across state lines may exacerbate owner-manager agency problems both by reducing the threat of a takeover and by limiting competition. Managers of banks in regulated states may invest in negative net-present value projects, either because they exert less effort in loan screening and monitoring or because they earn utility from some unproductive loans. After deregulation, these managers may be forced to increase shareholder value either to avoid takeovers or to protect against the loss of market share to more efficient competitors. According to this <u>disciplining hypothesis</u>, the banking system improves because managerial effort increases for all banks.

A. Measures of banking performance

To test for these conjectured effects of geographic restrictions, we first examine how aggregate measures of bank profitability and bank loan quality change following deregulation. We rely here on state-level aggregate data rather than bank-level data because we want to test for the possibility that, following deregulation, a selection mechanism weeded out less efficient banks. Such a conjecture cannot be tested easily by examining the effects of deregulation on individual banks since we would be restricted to a sample of surviving banks, which biases any tests. Aggregate data on the banking industry as a whole would capture any selection process as well as possible disciplining effects. We analyze bank-level data in the following section to test between the conjectured selection and disciplining hypotheses.

We measure profitability by return on assets (net income divided by total assets) and return on equity (net income divided by book value of equity). Net income, assets and book value of equity are aggregated to the state level by summing across all commercial banks operating in a given state. The dependent variable equals the total net income of all banks in a

state divided by the total assets (or equity) of all banks in that state. These data are taken from year-end Quarterly Reports of Condition for 1975 to 1992.

We examine four measures of loan quality. First, we use the fraction of total loans classified as "non-performing." End-of-year non-performing loan amounts for all banks over the 1982 to 1992 period are taken from *Quarterly Reports of Condition*. A state-level aggregate non-performing loan amount is derived by summing over all banks in each state. The final variable of interest is the ratio of non-performing loans to total loans held by all banks in each state.

Our second loan quality indicator is the fraction of loans written off during the year. Net charge-offs (gross charge-offs less recoveries) for individual banks are again taken from end-of-year Quarterly Reports of Condition over the 1976-1992 period. State-level total charge-offs are derived, and the dependent variable is the ratio of charge-offs to total loans.

Third, we use the flow of loan loss provisions during the current year divided by total loans, again aggregated up to the state level. Loan loss provisions, like non-performing loans and charge-offs, should be higher at banks with poorer loan quality.⁶

Finally, we look at loans to insiders as our last measure of loan quality. "Insider loans" are defined as extensions of credit to executive officers and principal shareholders. We presume here that such loans are potentially less productive than standard loans. Insider loans may also

⁵ All loans 90 days or more past due and nonaccrual loans are classified as non-performing loans.

⁶ Non-performing loans, loan charge-offs and loan loss provisions reflect different aspects of the quality of a bank's loan portfolio. Non-performing loans indicate the current status of loans, charge-offs reflect bad loans made in the past, and loan loss provisions indicate a bank's expectations of future loan losses.

serve as a proxy for the degree to which a bank is operated for the benefit of its management.

Data on insider loans are available from the Quarterly Reports of Condition beginning in 1983.

B. Changes in performance

We use the dates reported in Table 1 to construct two indicator variables equal to 1 for states permitting branching and interstate banking. We use these indicator variables to estimate the effects of the policy changes in the following pooled time series/cross-section model:

$$\begin{aligned} Perf_{t,i} &= \alpha_t + \beta_i + \gamma_1 Branch_{t,i} + \gamma_2 Bank_{t,i} \\ &+ \eta_1 PIG_{t,i} + \eta_2 PIG_{t-1,i} + \eta_3 PIG_{t-2,i} + \eta_4 PIG_{t-3,i} + \epsilon_{t,i} \end{aligned}$$

where $Perf_{t,i}$ equals one of the six measures of performance; $Branch_{t,i}$ is an indicator equal to 1 for states without restrictions on branching via M&A; $Bank_{t,i}$ is an indicator equal to 1 for states that have entered into an interstate banking agreement; $PIG_{t,i}$ is the growth rate in personal income.

In this specification, β_i measures the state-specific component of banking performance; α_i measures the national business cycle at time t; PIG removes the effects of the local (i.e. state-specific) business cycle; γ_1 and γ_2 measure the changes in performance stemming from the two types of deregulation.⁷

The results of the basic models outlined above appear in Table 2. The first two columns present the profitability results. These results show that profits, if anything, *increase* following deregulation. Following branching deregulation, we find that both return on assets and return on equity increase significantly. Following interstate banking, return on equity and return on assets

⁷ In constructing the deregulation indicators, we drop the year in which the deregulation went into effect. We also drop Delaware and South Dakota from the analysis entirely. These two states have passed laws providing tax incentives for credit card banks to operate there.

also rise, but the latter increase is not statistically significant. Note that the point estimates are also economically large. For instance, return on assets increases by 16 percent relative to the unconditional mean following branching deregulation.

The last four columns present the changes in bank loan performance. These results appear consistent with the profitability results. In particular, we find large improvements in all four measures of bank lending quality following branch deregulation. We also see smaller improvements following interstate banking, where two of the four loan performance measure improvements are statistically significant at the 10 percent level. The improvements in loan quality, along with the increases in profitability, suggest that banks are, on average, operating more efficiently following deregulation.⁸

C. Did deregulation cause the improvements?

A plausible explanation for the observed improvements in profitability and loan quality following branching deregulation is that states deregulated when their economies were doing poorly. Following deregulation, banks' loan portfolios may improve as the economies recover from the trough of the business cycle. The timing of the policy change may create a spurious association between branching deregulation and measures of profits and loan quality. This possibility is suggested by the fact that 25 of the 35 states that deregulated their branching restrictions during the sample period changed policy after 1984, the first of many years of

⁸ Shrinking non-performing loans, charge-offs and loan loss provisions need not reflect superior screening and monitoring of borrowers. Instead, they may reflect changes in the bank loan portfolio; banks may now be making fewer risky loans. We tested for this possibility by looking for changes in banks' loan portfolios after branch deregulation. We find that banks' loan portfolios show no movement away from the two riskiest loan categories, C&I loans and commercial real estate loans, following branch liberalization

dramatically increased bank failure rates. It is possible that, confronted with a severe negative shock to the economy and to the banking system, small banks -- the traditional constituency for branching restrictions -- dropped their opposition to branching in order to find higher purchase prices when exiting the distressed banking industry. Regulators may have pushed for liberalized branching to increase bank consolidation and to wean out weaker banks. 10

Another explanation for the observed improvements in bank performance following branching is that state legislatures anticipated significant growth in the states' economies and deregulated their banking systems so that banks were better able to fund attractive investment projects.

Neither of these explanations of the improved bank performance is plausible. By including contemporaneous as well as lagged values of personal income growth in our regression, we have controlled explicitly for state-specific business cycles. The estimated effect of branching and interstate banking on bank performance is not biased by any correlation between these deregulations and the business cycle. Moreover, Jayaratne and Strahan (1996) find that states were just as likely to deregulated branching restrictions during the downswing of the business cycle as during the upswing. Finally, Jayaratne and Strahan (1996) also establish that neither bank lending nor manufacturing investment increased following branch deregulation, which is inconsistent with the conjecture that states allowed branching in anticipation of an improved investment climate.

⁹ 1296 banks were subject to FDIC intervention over the nine year interval between 1984 and 1992. In contrast, a mere 25 banks failed over the nine years prior to 1984 [FDIC 1993].

¹⁰ We are grateful to Charles Calomiris and to Stavros Peristiani for suggesting this possibility.

Nevertheless, it is possible that the observed performance changes are the result of some external financial sector innovations not adequately captured by the personal income growth variable that both raised the cost of these geographical restrictions (and therefore raised the likelihood of their removal) and led to improved bank lending. To account for this possibility, we look to the timing of the changes associated with deregulation. If an unmeasured financial sector innovation led to bank deregulation and improved banking performance, then we should observe some of the performance improvements to occur shortly before (as well as after) deregulation (assuming that the political process operates slowly enough such that changes in bank performances stemming from the financial innovation occur before deregulation, which seems likely). If, on the other hand, deregulation was the source of the improvements in bank lending, all of the benefits should occur after the deregulation has gone into effect.

We examine the timing of the banking improvements by adding four additional indicator variables to our basic model. The first two indicators are equal to 1 during the first five-year window before branching deregulation and during the first five-year window after branching deregulation, and 0 otherwise. The second two indicators are equal to 1 during the first five-year window before interstate banking deregulation and during the first five-year window after interstate banking deregulation, and 0 otherwise. The model with windows is estimated as follows:

$$Perf_{t,i} = \alpha_{t} + \beta_{i} + \gamma_{1}Branch_{t,i} + \gamma_{2}Bank_{t,i} + \gamma_{3}Branch_{t,i}^{-5} + \gamma_{4}Branch_{t,i}^{+5} + \gamma_{5}Bank_{t,i}^{-5} + \gamma_{6}Bank_{t,i}^{+5} + \gamma_{1}PIG_{t,i} + \eta_{2}PIG_{t-1,i} + \eta_{3}PIG_{t-2,i} + \eta_{4}PIG_{t-3,i} + \epsilon_{t,i}$$

where $Branch^{-5}_{l,i}(Bank^{-5}_{l,i})$ equals an indicator equal to 1 during the five years leading up to branching (interstate banking) deregulation; and $Branch^{+5}_{l,i}(Bank^{+5}_{l,i})$ equals an indicator during

the five years immediately after deregulation. In this model, γ_3 equals the change in *Perf* during the five-year window leading up to branching deregulation relative to the years prior to that; $\gamma_1 + \gamma_4$ equals the change in *Perf* during the five year window after branching deregulation; and γ_1 equals the permanent change in *Perf* after deregulation (i.e. in years six and out).

These results are reported in Table 3. Looking first at branching deregulation, we see that all of the beneficial changes in our six performance measures occur *after* deregulation. The coefficient on the indicator for the five-year window prior to branching deregulation is never statistically significant. In other words, all of our measures of performance look the same during the five years before deregulation as they did six years or more before deregulation. By contrast, we find that non-performing loans, charge-offs and loan loss provisions all improve after branching deregulation, with most, but not all of the improvement occurring during the first five-year window after deregulation. The improvements appear to extend beyond five years, since the coefficients on the branching indicator (γ_1) are negative and statistically significant. The non-performing loans to total loans ratio, for instance, falls by 0.8 percentage points during the first five-year window, and falls by an additional 0.4 percentage points subsequently, for a decline of 1.2 percentage points five years after branching deregulation. ¹¹

For profitability, the coefficients on the two post-deregulation indicators for branching are not individually statistically significant. Nevertheless, the point estimates suggest higher profits

These results are in the spirit of Granger-Causality tests, although such tests usually include lags of the dependent variable on the right-hand side. We have estimated models similar to those in equation (2) with three lags of the dependent variable instead of the personal income growth variables. These results are qualitatively similar to those presented below and are available on request.

after branching deregulation, both during the first five years as well as for years six and out, confirming the results in Table 2.

Looking next at interstate banking, the story becomes somewhat less clear. Here, we find significant improvements in non-performing loans, charge-offs and loan loss provisions both five years before as well as five years after deregulation, while we find that the long-run effect is not statistically significant. This may mean that both the deregulation and the improvements in loan quality were caused by some unmeasured financial sector innovation, or it may mean that bank managers perform better in anticipation of an increased threat of takeover following interstate banking deregulation.

Overall, we remain agnostic on whether or not interstate banking is associated with increased profitability and improved loan quality. The results of Table 2, which do not account for timing, suggest that loan quality measures are somewhat improved on average after states enter an interstate banking agreement, although not always statistically significantly so. The weaker results for interstate banking may be due to the lack of power associated with the estimated coefficients on the interstate banking indicator; most states deregulated interstate banking restrictions around the middle of the 1980s. Even if bank performance improved following interstate banking, the results of Table 3 raise doubts about the direction of causality. Since three performance measures improve before interstate banking is allowed, we cannot reject the possibility that an unmeasured banking innovation improved bank performance and also persuaded state legislatures to allow interstate banking.

IV. How did deregulation improve bank performance?

We have shown that bank performance improves dramatically following intrastate branch deregulation. All of the benefits occur after deregulation, not before, suggesting that the branching policy change helped improve banking. Why did banks perform better after branching was permitted? The efficiency of the banking system may have improved because better-run, more efficient banks are able to gain market share over their less efficient competitors after deregulation (the selection hypothesis) or because managerial effort increases following deregulation (the disciplining hypothesis).¹²

To test these two hypotheses, we rely in this section primarily on banking company-level data. Unlike the statewide aggregate data used in the previous section that capture the combined effects of all changes in the banking system which may have affected bank performance following deregulation, banking company-level data allow us to distinguish between the selection and disciplining hypotheses.

following deregulation is that constraints on geographical expansion prevented banks from operating at the optimal scale. We discount this possibility, however, for three reasons. First, there is scant evidence of scale economies in banking, at least for banks with total assets above \$500 million (see Berger, Hunter and Timme, 1993). It is implausible that the large improvements that we have found in the state-level aggregates could be explained by inefficiently small banks moving closer to the optimal scale. In 1980, for instance, banks with under \$500 million in assets (in 1994 dollars) held less than 30 percent of total assets in the banking system. Second, although the market share of small banks has fallen over the past two decades, thousands of small banks remain in operation. Third, we have estimated the change in our performance measures following branching deregulation for small (banks with assets under \$100 million) and large banks separately in the same fixed effects model of Table 2 (not shown). We find that the improvements are *larger* for large banks than for small, a finding inconsistent with the economies of scale explanation. These results are available on request.

Both the selection and disciplining hypotheses imply that the variability of performance across banks should diminish following deregulation. Fewer high cost banks should be able to survive following deregulation, hence lowering cross-bank variability (selection). Moreover, banks with more severe management/shareholder agency problems should improve performance more than banks with less severe agency problems, again lowering variability (disciplining).

Table 4 reports evidence consistent with these implications.¹³ For each state, we construct the return on equity, return on assets, and the ratios of charge-offs, non-performing loans, loan loss provisions and insider loans to loans for all banking companies (independent banks and bank holding companies) operating in the state during the year just prior to branching deregulation and 5 years after deregulation.¹⁴ Since we have 32 states deregulating during different years, we first remove the state and year effects (i.e. means) from the variables. We present the standard deviation for the pre- and post-deregulation years in the table. In five of the six cases, the cross-sectional variability declines significantly after deregulation. Together with Table 3, these findings suggest that some banking companies were able to operate inefficiently (relative to other banking companies) prior to deregulation but were less able to do so after.

¹³ For this and subsequent analyses done at the banking company level, we focus on intrastate branching deregulation since the average performance improvements are much larger following this type of deregulation than following interstate banking deregulation.

¹⁴ Some states entered interstate banking agreements during the five-year window. For these states, we use the year just prior to the year in which the state entered the interstate banking agreement as the end of the window. We dropped four states that entered interstate banking agreements in the same year or one year after branching was deregulated (WV, TN, OR and NH).

The evidence in Table 4 does not help determine whether selection or disciplining is the key to the performance improvements. In the next two sub-sections, we provide evidence that both factors played a role.

A. Disciplining

The sharp decline in insider lending following branching deregulation provides prima facie evidence that disciplining is enhanced following branching deregulation (Table 2, column (6)). Of course, some insider lending is probably efficient (i.e. some insider loans have positive net present value), but some of that lending is likely to be an additional form of management compensation. Its decline suggests better disciplining of management. This fact, however, does not tell us how disciplining occurs. One standard solution to management/shareholder agency problems is the threat of takeover. But other means may exist to discipline management. For instance, the threat of entry and declines in market share may prod managers of banks in previously protected markets to exert more effort in minimizing costs and maximizing profits.

To test whether there is any role for disciplining from the takeover market, Table 5 reports estimates of the change in the acquisition rate following deregulation. Here, we measure the acquisition rate as the total dollar value of assets in banking companies acquired in a given year in a given state, divided by total assets in that state at the beginning of the year. We do not include acquisitions that required FDIC assistance. Nor do we include mergers and acquisitions that arise from corporate reorganizations, which involve no change in the control of banking

assets. For example, mergers among banks held by the same bank holding company are not included.¹⁵

As reported, the acquisition rate increases quite sharply following interstate banking deregulation but not following intrastate branch deregulation. It is not surprising, of course, that banks are acquired at an increased rate after a state joins an interstate banking agreement with other states. What is surprising is that there is no increase following branch deregulation. This finding casts some doubt on the hypothesis that disciplining is enhanced after branching deregulation through a more active takeover market. ¹⁶

Taking the results of Tables 4 and 5 together, we find a decline in performance variability but no increase in the rate of acquisitions following branching deregulation. In the absence of increased takeover activity, the decline in performance variability seems more consistent with disciplining than with selection. It suggests that some banks suffer more from lax management and that better disciplining improves the performance most for banks which suffer most from these agency problems. Since we see little change in the acquisition rate, these results suggest that better disciplining occurs primarily through enhanced competition in the goods market. That is, faced with the prospect of losing market share to new competitors once branching restrictions are lifted, lax managers may have improved their performance.

¹⁵ These data are taken from Rhoades (1986) and Rhoades (1996). We thank Stephen Rhoades for providing us the data.

¹⁶ Column (2) of Table 5 shows that the HHI increases sharply following branching deregulation, suggesting that there are important changes in the size distribution of banks, notwithstanding the acquisition result. This suggests that some banks grow following branching deregulation, either through internally generated growth (e.g. de novo branching) or through branch purchases.

Whether disciplining via the takeover market plays an important role after interstate banking remains unresolved. On the one hand, we see a sharp increase in takeovers following interstate banking and some small improvements in lending. Also, Hubbard and Palia (1995) find increases in CEO turnover and increases in the performance sensitivity of CEO pay after states enter interstate banking agreements. On the other hand, the improvements we do see following interstate banking seem to occur during the five years *before* deregulation. This is only consistent with the disciplining hypothesis if deregulation were anticipated and thereby spurred increased managerial effort during the five year pre-deregulation window.

On balance, the evidence here provides weak support for the disciplining hypothesis -- the idea that, by limiting competition in the goods market and possibly by limiting the takeover market, geographic restrictions weakened management discipline and raised banks' costs. 17

B. Selection

According to the selection hypothesis, average bank performance improves following branching deregulation because the poorly-run, inefficient banking companies lose ground to the better-run, more efficient banking companies. We conduct two exercises to test this notion.

First, we compare banks that grow following branching deregulation with banks that shrink. If the selection hypothesis is correct, banks that grow following deregulation should be a better class of banks than those that shrink. In this exercise, we separate banks into "growers" and "shrinkers" based on their growth rates after deregulation; we then compare their pre-

Prowse (1995) casts some doubt on the importance of bank takeover activity in disciplining bank managers. He finds that hostile takeovers almost never occur in banking, in sharp contrast to their frequency in nonfinancial industries. It seems reasonable that the threat of takeover can only discipline management effectively if some hostile takeovers actually take place.

deregulation performance. The second exercise is the reverse of the first. Bank are first separated by their pre-deregulation performance into high and low performers. The selection hypothesis predicts that the better performers should grow faster following deregulation.

The results of the first exercise are in Table 6. Here, we classify as "growers" all banking companies that grew faster than the median banking company during the six-year period beginning the year before deregulation and ending five years after branching deregulation. "Shrinkers" are those banking companies that either grew slower than the median bank or which exited altogether (through mergers or failures).

We find that for all of our performance measures, banking companies that grow have better ex-ante performance than banking companies that shrink.¹⁸ For example, during the post-deregulation period (columns 4-6), growers' mean return on equity was 1.2 percentage points below the median, while shrinkers' mean return on equity was 6.3 percentage points below the median.¹⁹ The median value for growers' return on equity was 0.81 percentage points above the unconditional median and 0.80 percent points below the unconditional median for shrinkers.

Return on equity exceeded the median 57 percent of the time for growers but only 44 percent of the time for shrinkers.²⁰

¹⁸ We do not include either insider lending or non-performing loans in this analysis, however, since these items are only available beginning in 1982 and 1983.

The mean return on equity is below the median for both growers and shrinkers because the unconditional mean return on equity is 3.7 percentage points below the unconditional median.

For each performance measure, we subtract the median of that measure which prevailed in that bank's state and that year, thereby eliminating the effects of the business cycle (which is required because we are pooling different states in different years).

This positive correlation between bank performance and asset growth, however, may not be due solely to a selection process. Banks facing better loan demand and investment opportunities are likely to show superior performance and faster asset growth. We account for this possibility by using the grower-shrinker comparison in the pre-deregulation period (Table 6, columns 1-3) as a "control." We observe the same pattern: growers are more profitable and have better loan quality, on average, than shrinkers in the pre-deregulation period. Nevertheless, the contrast between growers and shrinkers is sharper in the post-deregulation period (Table 6, columns 7 and 8). This suggests that the positive correlation between growth and bank performance after branching deregulation is not entirely due to cross-sectional differences in demand conditions (assuming that such demand conditions generate the same degree of correlation between asset growth and bank performance before and after branching is allowed).

The exercise in Table 6 is limited by the fact that even if the *average* bank that grew after branching deregulation was a high-profit bank, a few, poorly-performing banks may have grown the most, and the results in Table 6 will not reveal that fact. This possibility, which undermines the selection hypothesis, is tested in Table 7 by reversing the exercise in Table 6. In Table 7, we ask whether ex-ante performance has more power to predict growth in subsequent years once branching is permitted. We separate bank into high performers and low performers based on return on assets, return on equity, net charge-offs and loan loss provisions. High performers are those banks that are above the median in profitability and below the median in net charge-offs

²¹ In the pre-deregulation period, growers are banking companies that grew faster than the median banking company during the period beginning seven years before deregulation and ending the year before deregulation. We use the same window width in the pre-deregulation period as in the post-deregulation period for those states where the post-deregulation window was less than six years (see note 14).

and loan loss provisions.²² We then regress banking company asset growth on each ex-ante performance indicator along with bank size and the capital-to-asset ratio at the beginning of the period.²³ (Bank size is included because large banks may be expected to grow relatively slowly, and capital is included because regulatory capital adequacy standards may constrain growth.)

As in Table 6, we report these regressions in both the pre- and post-deregulation periods. We find that in three of the four cases, the impact of performance on subsequent growth increases after deregulations (Table 7). For instance, before deregulation, banking companies with above-median return on equity grow 0.9 percentage points faster than banking companies with below-median return on equity. By contrast, once branching is allowed, this growth differential triples (to 3.3 percentage points) and become statistically significant. This pattern is also evident for net charge-offs and for loan loss provisions, but not for return on assets.

To test the statistical significance of the observed increase in the performance-growth association after deregulation, we pool the data from the pre- and post-deregulation periods. We then re-estimate the model with an added indicator variable for deregulation along with its interaction with each performance measure (Table 7, column 3). The model was estimated on the first differences of the growth, size, capital and performance indicator variables. This is equivalent to a fixed effects estimator, which is preferred here because the presence of the same

The median performance measures are conditional on year and state. For example, a bank in Alabama would be classified as an above-median ROE bank in 1980 if its ROE is above the median ROE in Alabama in 1980. This approach removes the effects of inflation and state-specific factors on profitability.

The relationships between the continuous measures of performance and growth were non-linear in both the pre- and post-deregulation periods. We therefore use an indicator variable of performance rather than the continuous variable so that comparisons of the coefficients across time are more easily interpreted.

banks in both periods is likely to induce a correlation between the error term in the two cross-sectional regressions. Table 7, column 3 shows that all four performance measures are statistically significantly better predictors of growth after branching than before branching.²⁴

The comparisons of average growth rates of high and low performance banks in Table 7 exclude banks that exit the industry due to failures or mergers. Such exits are excluded because their "growth rate" cannot be observed. Exiting banks, however, may carry significant information about any selection mechanism. To account for this, we estimate the probability that banking companies with above-median performance before deregulation exit the industry after deregulation. We compare this exit probability with that for banking companies that perform below the state median before branching was allowed. We find that each of the four performance measures predicts that better-performing banks have a lower probability of exiting than underperforming banks. Although we find a similar pattern for the period immediately before branching deregulation, the difference in exit probabilities between high and low performance banks increases once branching is permitted, further supporting the selection hypothesis.

For example, in the pre-deregulation period, a banking company with above-average return on equity had a 22 percent chance of exiting, while a banking company with below-average return on equity had a 26 percent chance of exiting -- the small difference in probabilities reflects a relatively weak selection mechanism. By contrast, after branching deregulation, a

²⁴ The pooled regression does not entirely replicate the cross sectional regressions, perhaps because they do not use quite the same data. The pooled regression is estimated using only those banking companies that survived the entire sample period. In contrast, the prederegulation cross-sectional regression is estimated on all banks that are alive over the prederegulation period, and this includes banks that exited once branching was allowed.

banking company with above-average return on equity had a 28 percent chance of exiting while a banking company with below-average return on equity had a 37 percent chance of exiting.²⁵

The results in Tables 6 and 7, along with the results on exit probabilities, suggest that selection effects became more powerful following branching deregulation. To what extent can this selection process explain the improvements in aggregate, state-level banking performance observed in Table 2? For example, Tables 2 and 3 suggest that the asset-weighted average of banking company return on equity increased by 9 percentage points after statewide branching was permitted, with 5 of the 9 percentage point increase occurring during the first five years after deregulation. What fraction of this improvement can be explained by the selection process?

We answer this question by answering the following question: what would be the asset-weighted average return on equity in the year before deregulation if the asset weights are based on the distribution of assets observed five years after deregulation? If, as the selection hypothesis predicts, the initially high-profit banks expand the most following deregulation, then the asset-weighted average return on equity based on bank sizes observed five years after deregulation should be greater than the asset-weighted return on equity based on bank sizes observed the year before deregulation.²⁶

This prediction of the selection conjecture is confirmed. The weighted average return on equity before deregulation was 8.3 percent. Based on the asset distribution observed five years later, the average increases to 9.3 percent. By contrast, in the pre-deregulation period we find

²⁵ These results are not shown in a table but are available on request.

²⁶ By using the same performance data for both averages, we hold constant other factors (such as disciplining) that may have improved performance following deregulation.

that the actual weighted average return on equity is *lower* than the weighted average based on assets observed five years later, falling from 14.8 percent to 11.7 percent. Using a difference-in-differences approach, these estimates suggest that selection effects could account for an increase in return on equity of about 4 percentage points ([9.3-8.3]-[11.7-14.8]). This is below the estimated increase in Table 2 (although it is within two standard errors of that estimate) but it accounts for 80 percent of the increase in the return on equity observed within the first five years after branching (Table 3, column 2). The same simulation exercise yields an improvement of 50 percent or more of the estimated effects of the other three performance measures (loan loss provisions, return on assets and charge-offs) estimated in Table 2.²⁷

To summarize, banks that have higher profits and better loan portfolios before branching deregulation are observed to grow at the expense of other banks once branching is permitted.

Moreover, banks with initially poor performance exit the industry more frequently. Although similar patterns are evident in the period before deregulation, the performance-growth relation is stronger after branching deregulation, consistent with the selection hypothesis. The simulation exercise suggests that selection effects account for most of the estimated improvements in bank performance.

V. Conclusions

The literature on the effects of entry regulation on efficiency has shown that such interventions can reduce static efficiency by preventing firms from allocating their assets

²⁷ These results are based on data pooled across all states. We have also done these simulations for each state separately and found similar results. For instance, in 20 out of 31 cases, we find that re-weighting the average return on equity based on the size distribution five years after deregulation leads to an increase in this average, as predicted by the selection hypothesis.

optimally. For example, trucking regulations prevented carriers from hauling regulated commodities on return trips, making empty backhauls a serious problem. Airline regulations increased costs by reducing load factors and by forcing airlines into serving inefficient routes to small communities. However, relatively little has been done on the effects of entry regulation on dynamic efficiency.

In this paper, we find evidence that long-standing branching restrictions in banking served as entry barriers that prevented more efficient banks from expanding at the expense of their less efficient rivals. By retarding the "natural" evolution of the industry, such restrictions reduced the performance of the average banking asset.

The literature on price and entry regulations suggests that these regulations can also enhance market power, and previous work in banking has found that restrictions on the ability of banks to branch statewide or expand across state lines had similar effects. However, welfare losses associated with such regulation-induced market power in banking are probably small (Rhoades 1982). We do not estimate the welfare gains associated with the observed improvements in loan portfolios following branching deregulation. Nevertheless, the large improvements in loan portfolios (30 percent or more) after statewide branching suggest that entry restrictions exacted significant welfare losses.

The improvements in efficiency of the banking system suggests that benefits may have flowed to both bank owners and bank customers -- depositors and borrowers. As noted, previous work both in banking and in other industries suggests that restrictions on entry and expansion may enhance market power. To the extent that market power was reduced following deregulation, depositors and borrowers may therefore have been made better off both because

costs fell and because prices moved closer to costs. On the other hand, if market power was not diminished, the benefits of the reduced costs could have been shared between owners (in the form of higher profits), depositors and borrowers. Our finding that profits increased following deregulation may be more consistent with this latter view, although the long-run effects of deregulation on profitability are not clear from our findings since most of the states deregulated after 1984. An analysis of the effects of deregulation on market power in banking would require looking carefully at changes in deposit and loan interest rates. We leave this question for future research.

References

- Amel, Dean, "State Laws Affecting the Geographic Expansion of Commercial Banks," manuscript, Board of Governors of the Federal Reserve System, 1993.
- Amel, Dean and Nellie Liang, "The Relationship between Entry into Banking Markets and Changes in Legal Restrictions on Entry," *The Antitrust Bulletin*, XXXVII(1992), 631-649.
- Berger, Allen N., William C. Hunter and Stephen G. Timme, "The Efficiency of Financial Institutions: A Review and Preview of Research Past, Present and Future," *Journal of Banking and Finance* 17(1993) 221-249.
- Berger, Allen N., Anil K Kashyap and Joseph M. Scalise, "The Transformation of the U.S. Banking Industry: What a Long Strange Trip It's Been," *Brookings Paper on Economic Activity*, II(1995), 55-218.
- Berger, Allen and Timothy Hannan, "The Price-Concentration Relationship in Banking," Review of Economics and Statistics, CXXI (1989), 291-299.
- Calem, Paul, "The Impact of Geographic Deregulation on Small Banks," Business Review, Federal Reserve Bank of Philadelphia (December 1994).
- Calomiris, Charles, "Regulation, Industrial Structure, and Instability in U.S. Banking: An Historical Perspective," in Michael Klausner and Lawrence White (eds.) Structural Change in Banking, New York University: New York 1993, 19-116.
- Demsetz, Rebecca S. and Philip E. Strahan, "Size, Diversification and Risk at U.S. Bank Holding Companies," *Journal of Money, Credit and Banking*, forthcoming.
- Economides, Nicholas, R. Glenn Hubbard and Darius Palia, "The Political Economy of Branching Restrictions and Deposit Insurance: A Model of Monopolistic Competition among Small and Large Banks," NBER working paper No. 5210(1995).
- Evanoff, Douglas and Diana Fortier, "Reevaluation of the Structure-Conduct-Performance Paradigm in Banking," *Journal of Financial Research*, I (June 1988), 277-294.
- Flannery, Mark, "The Social Costs of Unit Banking Restrictions," *Journal of Monetary Economics*, 13(1984), 237-249.
- Federal Deposit Insurance Corporation, *Historical Statistics on Banking*, (Washington, D.C., 1993).
- Hsiao, Cheng, Analysis of Panel Data, (Cambridge: Cambridge University Press, 1986).

- Hubbard, Glenn and Darius Palia, "Executive Pay and Performance: Evidence from the U.S. Banking Industry," *Journal of Financial Economics*, XXXIX(1995), 105-130.
- Jayaratne, Jith and Philip E. Strahan, "The Finance-Growth Nexus: Evidence from Bank Branch Deregulation," Quarterly Journal of Economics, CXI(1996).
- McLaughlin, Susan, "The Impact of Interstate Banking and Branching Reform: Evidence from the States," Current Issues in Economics and Finance, I(1995), Federal Reserve Bank of New York.
- Moore, Robert M. "Does Geographic Liberalization Really Hurt Small Banks?" Financial Industry Studies, Federal Reserve Bank of Dallas, December 1995.
- Prowse, Stephen, "Alternative Methods of Corporate Control in Commercial Banks," mimeo, May 26, 1996.
- Rhoades, Stephen, "Welfare Loss, Redistribution Effects and Restriction of Output Due to Monopoly in Banking," *Journal of Monetary Economics* 9(1982), 375-387.
- Rhoades, Stephen, "The Operating Performance of Acquired Firms in Banking before and after Acquisitions," Federal Reserve Board of Governors Staff Study No. 149, 1986.
- Rhoades, Stephen, "Bank Mergers and Industrywide Structure, 1980-94," Federal Reserve Board of Governors Staff Study No. 169, 1996.
- Schranz, Mary, "Takeovers Improve Firm Performance: Evidence from the Banking Industry," *Journal of Political Economy*, CI(1993), 299-326.
- White, Halbert, "A Heteroskedasticity-consistent covariance Matrix Estimator and A Direct Test for Heteroskedasticity," *Econometrica*, XLVIII(1980), 817-830.
- Winston, Clifford, "Economic Deregulation: Days of Reckoning for Microeconomists," *Journal of Economic Literature*, XXXI(1993), 1263-1289.

Table 1

Deregulation of Restrictions on Geographical Expansion, by State

AK AL AL 1981 1987 AR AR Not Deregulated by 1992 1988 AZ Deregulated Before 1970 1986 CA Deregulated Before 1970 1987 CO 1991 1988 CO 1991 1988 CO 1991 1988 CO 1989 CO 1989 CO 1989 CO 1989 CO 1989 CO 1989 CO 1988 1985 CO 1988 1986 CO 1988 CO 1988 1986 CO 1988		Intrastate Branching Allowed by Merger or Acquisition (1)	State Entered an Interstate Banking Agreement (2)
AL 1981 1987 AR Not Deregulated by 1992 1989 AZ Deregulated Before 1970 1986 CA Deregulated Before 1970 1987 CO 1991 1988 CT 1980 1983 DC Deregulated Before 1970 1988 CT 1988 1985 GA 1988 1985 GA 1988 1985 GA 1986 Not Deregulated by 1992 IA Not Deregulated by 1992 1985 IL 1988 1986 IN 1989 1986 KS 1987 1992 KY 1990 1988 IA 1987 1992 KY 1990 1988 IA 1988 1987 MA 1984 1985 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1997 1996 MM Not Deregulated by 1992 1986 MN Not Deregulated Before 1970 1985 ME 1975 1978 MI 1990 1986 MN Not Deregulated Before 1970 1986 MN Not Deregulated Before 1970 1986 MN 1990 1988 MT 1990 1986 NS 1988 1988 MT 1990 1986 NS 1988 1988 MT 1990 1986 NS 1988 1988 MT 1990 1985 NC Deregulated Before 1970 1985 NF 1997 1991 NF 1987 1991 NF 1988 1987 NF 1996 NF 1997 1985 NF 1998 NF 1998 NF 1998 NF 1999 1985 NF 1996 NF 1998 1985 NF 1998 NF 1999 1985 NF 1998	AK	. Deregulated Before 1970	1982
AR Not Deregulated by 1992 1988 AZ Deregulated Before 1970 1986 CA Deregulated Before 1970 1987 CO 1991 1988 CT 1991 1988 DC Deregulated Effore 1970 1983 DC Deregulated Before 1970 1985 FL 1988 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 1985 IA Not Deregulated by 1992 1991 ID Deregulated Before 1970 1985 IL 1998 1986 KS 1989 1988 IN 1989 1988 IN 1988 1987 MA 1988 1987 MM 1997 1985 MM 1997 1985 MM Not Deregulated Before 1970 1985 MM 1989 1986 MN Not Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1988 MN Not Deregulated Before 1970 1985 ME 1987 1986 MN Not Deregulated Before 1970 1985 ME 1987 1986 MN Not Deregulated Before 1970 1985 ME 1987 1986 MN 1988 1987 1986 MN Not Deregulated Before 1970 1985 ND 1980 1986 MS 1986 1988 MT 1990 1986 MS 1986 1988 MT 1990 1986 NS 1986 1988 MT 1990 1986 NS 1986 1988 MT 1990 1986 NS 1986 1988 MT 1990 1989 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1986 ND 1997 1986 ND 1997 1986 ND 1997 1986 ND 1997 1986 ND 1998 1999 NF 1987 1988 T 1988 1987 NJ 1991 1988 NT 1998 1987 NJ 1991 1988 NT 1998 1987 NT 1988 NA 1985 1986 NT 1989 1985 NT 1988 1987 NT 1988 1988 NT 1989 1989 NT	AL	~	1987
AZ Deregulated Before 1970 1986 CA Deregulated Before 1970 1987 CO 1991 1998 CT 1980 1983 DC Deregulated Before 1970 1985 FL 1988 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated by 1992 1991 IA Not Deregulated Before 1970 1985 IL 1988 1986 IN 1988 1986 IN 1989 1986 IN 1988 1986 IN 1988 1986 IN 1989 1986 IN 1988 1986 IN 1980 1986 IN 1988 1987 IN 1985 1986 IN 1986 1986 IN 1986 1986 IN 1986 1986		Not Deregulated by 1992	1989
CO 1991 1988 CT 1980 1983 DC Deregulated Before 1970 1985 FL 1988 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated Before 1970 1985 IA Not Deregulated Before 1970 1985 IL 1988 1986 IN 1989 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 KY 1990 1983 MA 1984 1983 MD 1987 1985 ME 1975 1978 MI 1987 1986 MN Not Deregulated Before 1970 1985 MS 1986 1988 MT 1990 1983 NC Deregulated Before 1970 1985			1986
CT 1980 1983 DC Deregulated Before 1970 1985 FL 1983 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated Before 1970 1985 IA Not Deregulated Before 1970 1985 II 1988 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MN Not Deregulated Before 1970 1985 MS 1986 1988 MT 1987 1991 NC Deregulated Before 1970 1985 NH 1987 1991	CA	Deregulated Before 1970	1987
DC Deregulated Before 1970 1985 FL 1983 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated Before 1970 1985 IL 1988 1986 IN 1988 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1987 MA 1984 1983 MD Deregulated Before 1970 1985 MI 1987 1986 MN Not Deregulated Before 1992 1986 MS 1986 1988 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1986 ND 1987 1987 ND 1985 1986		1991	1988
FL 1986 1985 GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated Before 1970 1985 IL 1988 1986 IN 1989 1986 IN 1989 1986 IN 1989 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1987 1986 MN Not Deregulated by 1992 1986 MS 1986 1986 MS 1986 1986 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 ND 1987 1986 ND 1985 1980 ND			
GA 1983 1985 HI 1986 Not Deregulated by 1992 IA Not Deregulated by 1992 1991 ID Deregulated Before 1970 1985 IL 1988 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MO 1980 1986 MS 1980 1986 MS 1980 1986 MS 1980 1988 MT 1990 1996 MS 1986 1988 MT 1990 1991 ND 1987 1991 ND 1987 1991 NB 1987 1981 NM 1987			
HI 1986 Not Deregulated by 1992 IA Not Deregulated by 1992 1991 ID Deregulated Before 1970 1985 IL 1988 1996 IN 1988 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1996 MN Not Deregulated Before 1970 1985 MS 1986 1986 MS 1986 1988 MT 1990 1983 NC Deregulated Before 1970 1985 ND 1987 1991 ND 1987 1991 NL 1985 1990 NH 1987 1981 NJ 1987 1986 NJ 1987 1986			-
A			
Deregulated Before 1970 1985 IL			
IL 1988 1986 IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1986 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1988 MT 1990 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1986 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1985 OR 1985 1986 PA 1982 1986 PA			
IN 1989 1986 KS 1987 1992 KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1996 MN Not Deregulated by 1992 1986 MS 1986 1988 MT 1990 1985 NC Deregulated Before 1970 1985 ND 1987 1991 NH 1987 1991 NH 1987 1991 NH 1987 1991 NF 1988 1989 NV Deregulated Before 1970 1985 NY 1976 1985 NY 1978 1986 NY 1988 1998 NY 1988 1988 TH 1990 NH 1988 1989 NY 1978 1985 NY 1988 1988 TH 1989 NY 1978 1988 TH 1989 NY 1978 1985 NY 1988 1987 NF 1988 NF 1988 1987 NF 1988 1988 NF 1988 1987 NF 1988 1987 NF 1988 1987 NF 1988 1988 NF 1989 1989 NF 1989 1989 NF 1989 1988			
KS 1987 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1976 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1991 NE 1985 1990 NH 1987 1986 NM 1991 1987 NO Deregulated Before 1970 1985 NO 1985 NO 1986 NH 1987 1987 NO 1988 NO 1991 1987 NO 1988 NO 1991 1988 NO 1991 1988 NO 1991 1985 NO 1985 NO 1985 NO 1986 NO 1991 1985 NO 1985 NO 1986 NO 1988 1987 NO 1985 NO 1986 NO 1985 NO 1985 NO 1986 NO 1985 NO 1986 NO 1988 NO 1987 NO 1988			
KY 1990 1984 LA 1988 1987 MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1987 1991 NE 1987 1990 NH 1987 1991 NJ 1977 1986 NM 1991 1985 NY 1976 1985 NY 1976 1985 NY 1970 1985 PA 1985 1986 PA 1985 1986 PA 1985 1986 PA 1985 1986 PA 1986 1985 TX 1988			
LA 1988 1987 MA 1984 1983 MD Dergulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MN NOT Deregulated by 1992 1986 MN NOT Deregulated by 1992 1986 MS 1986 1988 MT 1990 1993 NC Dergulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1991 NF 1987 1991 NF 1987 1995 NN 1997 1985 NN 1991 1985 NN 1988 1987 NN 1988 1986 PA 1982 1986 PA 1982 1986 NN 1988 1987 NN 1988 1987 NN 1988 1987 NN 1988 1986 NN 1988 1987 NN 1988 1986 NN 1988 1987 NN 1988 1986 NN 1989 NN 1989 1985 NN 1988 1986 NN 1988 1986 NN 1988 1986 NN 1988 1986 NN 1988 1987 NN 1988 1988 NN 1988 NN 1988 1988 NN 1988 NN 1988 1988 NN 1989 NN 1989 NN 1987			
MA 1984 1983 MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1986 NW 1991 1985 NY 1976 1985 NY 1976 1985 NY 1979 1985 OR 1985 1987 OR 1985 1986 PA 1982 1986 PA 1982 1986 RI Deregulated Before 1970 1986 TX 1988 1987 UT 1981 1985 VT			
MD Deregulated Before 1970 1985 ME 1975 1978 MI 1987 1986 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1989 NY 1970 1985 NY 1970 1985 OK 1988 1987 OK 1988 1987 OR 1985 1986 RI Deregulated Before 1970 1986 PA 1982 1986 RI Deregulated Before 1970 1986 TX 1988 1987 UT 1981 1985 VT 1988 1985			
ME 1975 1978 MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1986 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1985 1986 PA 1985 1986 RI Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1980 1985 VT			
MI 1987 1986 MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1986 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 RI Deregulated Before 1970 1986 RI Deregulated Before 1970 1986 TX 1988 1987 UT 1988 1987 VA 1978 1985 VT 1981 1985 VT 1970 1988 WA 1985 1987 WV 1987 1988			
MN Not Deregulated by 1992 1986 MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1986 NM 1991 1987 NV 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1985 OK 1988 1987 OR 1985 1986 RI Deregulated Before 1970 1986 RI Deregulated Before 1970 1986 TN 1985 TX 1988 1986 TN 1985 TX 1988 1987 VT 1988 1985 TX 1988 1987 VT 1988 1985 VT 1989 1985 WA 1985 1986 WA 1985 1987 WV 1987		1-1-	
MO 1990 1986 MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1985 OK 1988 1987 OK 1985 1986 PA 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1985 VT 1981 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			
MS 1986 1988 MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1999 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 RI Deregulated Before 1970 1986 RI Deregulated Before 1970 1986 TN 1985 TX 1988 1987 UT 1988 1987 UT 1981 1988 VA 1976 1985 VY 1978 1985 VY 1988 1987 UT 1981 1984 VA 1988 1987 WV 1987 1985 WA 1985 1986 WA 1985 1986 WA 1985 1986 WA 1988 1987 WV 1987			
MT 1990 1993 NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TX 1988 1987 UT 1981 1985 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			
NC Deregulated Before 1970 1985 ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TX 1988 1987 UT 1981 1985 VA 1978 1985 VT 1970 1988 WA 1995 1987 WI 1990 1987 WV 1987 1988			
ND 1987 1991 NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			* * *
NE 1985 1990 NH 1987 1987 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	. =		
NH 1987 1987 NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1987 VA 1970 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987			
NJ 1977 1986 NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1987 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987			
NM 1991 1989 NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			
NV Deregulated Before 1970 1985 NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	·		
NY 1976 1982 OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1987 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987			
OH 1979 1985 OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			1982
OK 1988 1987 OR 1985 1986 PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988			1985
PA 1982 1986 RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988		1988	1987
RI Deregulated Before 1970 1984 SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988		1985	1986
SC Deregulated Before 1970 1986 TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	PA	1982	1986
TN 1985 1985 TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	RI	Deregulated Before 1970	1984
TX 1988 1987 UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	SC	Deregulated Before 1970	
UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	TN	1985	1985
UT 1981 1984 VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	TX		
VA 1978 1985 VT 1970 1988 WA 1985 1987 WI 1990 1987 WV 1987 1988	UT		
WA 1985 1987 WI 1990 1987 WV 1987 1988	VA		
WI 1990 1987 WV 1987 1988			
WV 1987 1988			
1			
WY 1988 1987			
	WY	1988	1987

Source: Amel (1993).

Notes: Delaware and South Dakota are dropped from the analysis since these two states have been dominated by credit card banks since the early 1980s.

Table 2

Changes in Bank Lending Quality Following intrastate Branch and Interstate Bank Reform

	Net Income/ Total Assets _(ROA) (1)	Net Income/ Equity _(ROE) _(2)	Non- Performing Loans/ Total Loans (3)	Net Loan Charge-offs/ Total Loans (4)	Loan Loss Provisions/ Total Loans (5)	Insider Loans/ <u>Total Loans</u> (6)
Intrastate Branching Indicator	0.0012	0.0943	-0.0081	-0.0045	-0.0048	-0.0014
	(2.10)*	(2.48)*	(3.68)*	(5.82) *	(5.35) *	(3.15) *
Interstate Banking Indicator	0.0007	0.0789	-0.0035	-0.0013	-0.0011	-0.0003
	(1.13)	(2.11) *	(1.74)	(1.84)	(1.16)	(0.67)
State Personal Income Growth	0.0425	1.0420	-0.2386	-0.0528	-0.0668	-0.0209
	(4.40) *	(2.85) *	(5.21)*	(5.31) *	(4.98) *	(2.74) *
Personal Income Growth (Lagged 1 Period)	0.0419	1.0805	-0.2194	-0.0541	-0.0509	-0.0089
	(3.87)*	(3.23) *	(5.73) *	(3.67) *	(3.74) *	(1.15)
Personal Income Growth (Lagged 2 Periods)	0.0250	0.5927	-0.1400	-0.0446	-0.0301	-0.0091
	(3.67)*	(1.68)	(3.48) *	(3.37) *	(3.32) *	(1.04)
Personal Income Growth (Lagged 3 Periods)	0.0147	-0.2391	-0.0385	-0.0236	-0.0171	0.0144
	(1.47)	(0.57)	(0.94)	(2.36) *	(1.23)	(2.37) *
Adjusted R-Squared	30.2%	11.8%	51.3%	44.1%	43.3%	69.5%
Sample Period	1975-1992	1975-1992	1982-1992	1976-1992	1975-1992	1983-1992
N	802	802	468	754	802	422
Dependent Variable Mean	0.77%	14.60%	2.67%	0.72%	0.86%	0.45%

Sources: Reports of Income and Condition. Dates for deregulation are from Amel (1993); see Table 1.

Notes:

Each column presents a pooled time series/cross-section regression with state level data for the sample period indicated. Each regression contains state and time fixed effects. Absolute value of t-statistic, based on White standard error, reported below each coefficient in parentheses; "" means significance at the 5% level. Delaware and South Dakota are dropped from each regression.

Table 3

Changes in Bank Lending Quality Following Intrastate Branch and Interstate Bank Reform, with Windows

	Net Income/ Total Assets (ROA) (1)	Net Income/ Equity (ROE) (2)	Non- Performing Loans/ <u>Total Loans</u> (3)	Net Loan Charge-offs/ <u>Total Loans</u> (4)	Loan Loss Provisions/ Total Loans (5)	Insider Loans/ <u>Total Loans</u> (6)
Intrastate Branching Indicator	0.0011	0.1049	-0.0123	-0.0058	-0.0054	-0.0009
	(0.99)	(1.35)	(3.01) *	(4.01)*	(3.31) *	(1.08)
Interstate Banking Indicator	-0.0005	-0.1722	-0.0002	0.0016	0.0003	0.0006
	(0.27)	(1.43)	(0.04)	(0.75)	(0.11)	(0.50)
5 Years Before Branching (0,1)	0.0001	-0.0288	-0.0021	-0.0002	-0.0005	0.0004
	(0.18)	(1.45)	(1.17)	(0.46)	(0.76)	(0.88)
5 Years After Branching (0,1)	-0.0003	-0.0545	0.0041	0.0022	0.0012	-0.0003
	(0.37)	(1.14)	(1.87)	(2.56) *	(1.25)	(0.59)
5 Years Before Banking (0,1)	0.0010	-0.0410	-0.0065	-0.0026	-0.0029	0.0013
	(1.78)	(1.67)	(2.95) *	(3.87) *	(3.68) *	(2.06) *
5 Years After Banking (0,1)	0.0021	0.1761	-0.0092	-0.0053	-0.0043	0.0005
	(1.44)	(1.90)	(2.60) *	(2.99) *	(2.34) *	(0.90)
State Personal Income Growth	0.0430	0.9491	-0.2293	-0.0550	-0.0690	-0.0222
	(4.23) *	(2.58) *	(5.08) *	(5.64) *	(4.97) *	(2.89) *
Personal Income Growth (Lagged 1 Period)	0.0427	0.9467	-0.2205	-0.0562	-0.0539	-0.0095
	(3.78) *	(2.72)*	(5.97)*	(3.89) *	(3.86) *	(1.27)
Personal Income Growth (Lagged 2 Periods)	0.0256	0.5549	-0.1403	-0.0461	-0.0315	-0.0082
	(3.79)*	(1.59)	(3.68)*	(3.49) *	(3.54) *	(0.94)
Personal Income Growth (Lagged 3 Periods)	0.0150	-0.2635	-0.0352	-0.0238	-0.0177	0.0140
	(1.50)	(0.63)	(0.87)	(2.46) *	(1.30)	(2.28) *
Adjusted R-Squared	30.9%	13.0%	53.5%	47.8%	45.3%	69.6%
Sample Period	1975-1992	1975-1992	1982-1992	1976-1992	1975-1992	1983-1992
N	802	802	468	754	802	422
Dependent Variable Mean	0.77%	14.60%	2.67%	0.72%	0.86%	0.45%

Sources: Reports of Income and Condition. Dates for deregulation are from Amel (1993); see Table 1.

Notes:

Each column presents a pooled time series/cross-section regression with state level data for the sample period indicated. Each regression contains state and time fixed effects. Absolute value of t-statistic, based on White standard error, reported below each coefficient in parentheses; " means significance at the 5% level . Delaware and South Dakota are dropped from each regression.

Table 4
<u>Cross Sectional Variability in Bank Performance Measure Before and After Branching Deregulation</u>

	Cross S Standard	ectional Deviation	
	Pre <u>Deregulation</u> (1)	Post <u>Deregulation</u> (2)	F <u>Statistic</u> (3) [H0: (1)=(2)]
Net income/assets	0.0160	0.0105	2.34 *
Net income/capital	0.1820	0.1122	2.63 *
Net charge-offs/loans	0.0198	0.0118	2.81 *
Loan loss provisions/loans	0.0214	0.0129	2.76 *
Non-Performing loans/loans	0.0360	0.0207	3.04 *
Insider loans/loans	0.0273	0.0289	0.90

Sources:

Reports of Income and Condition. Dates for deregulation are from Amel

(1993); see Table 1.

Notes:

Includes all states that deregulated their branching restrictions between 1970 and 1992. Banking companies with capital-asset ratios under 3 percent are dropped from the

analysis of return on capital. State and time specific means are removed from the data. *'

means significance at the 5% level .

Table 5 Changes in Banking Market Structure Following Deregulation

	Acquisition Rate (1)	Log of <u>State HHI</u> (2)
Intrastate Branching Indicator	-0.0064 (0.47)	0.3072 (6.49) *
Interstate Banking Indicator	0.0459 (2.48) *	0.0107 (0.14)
5 Years Before Branching (0,1)	0.0172 (1.67)	0.0344 (1.40)
5 Years After Branching (0,1)	0.0062 (1.21)	-0.1168 (3.62)*
5 Years Before Banking (0,1)	-0.0252 (1.87)	0.0629 (2.47) *
5 Years After Banking (0,1)	0.0001 (0.02)	0.0012 (0.03)
State Personal Income Growth	0.0959 (2.00) *	
Personal Income Growth (Lagged 1 Period)	-0.0058 (0.10)	
Personal Income Growth (Lagged 2 Periods)	-0.0095 (0.24)	
Personal Income Growth (Lagged 3 Periods)	0.0779 (1.66)	
Adjusted R-Squared Sample Period N Dependent Variable Mean	15.2% 1975-1992 802 2.2%	94.8% 1975-1992 802 7.46

Sources:

HHI is based on data from the Summary of Deposits; acquisition rates are based on data in Rhoades (1995).

Dates for deregulation are from Amel (1993); see Table 1.

Notes:

HHI equals the sum of the squared deposit shares held by all banking companies in the state. Banks under common ownership are consolidated. The acquisition rate equals the total dollar value of assets in acquired banks divided by total assets in all banks, by state and year. Each column presents a pooled time series/cross-section regression with state level data for the sample period indicated. Each regression contains state and time fixed effects. Absolute value of t-statistic, based on White standard error, reported below each coefficient in parentheses; "" means significance at the 5% level . Delaware and South Dakota are dropped from each regression.

Table 6
Ex-Ante Performance Measures for Growing and Shrinking Banks

	Pre-D Shrinkers (1)	eregulation Growers (2)	Period Difference (3) (2)-(1)	Post-I Shrinkers (4)	Deregulation Growers (5)	Period Difference (6) (5)-(4)	Difference in Differences (7) (6)-(3)	T-stat (8)
Net income/assets								
Mean	-0.19%	0.01%	0.20%	-0.60%	-0.06%	0.54%	0.34%	8.50
Standard Error	0.02%	0.01%		0.03%	0.02%			
Median	-0.05%	0.05%	0.10%	-0.09%	0.08%	0.18%	0.08%	
Percent above median	45.38%	54.43%	9.05%	42.43%	57.34%	14.91%	5.86%	
			•					
Net income/capital								
Mean _	-2.41%	-0.71%	1.70%	-6.28%	-1.23%	5.05%	3.35%	7.07
Standard Error	0.23%	0.15%		0.33%	0.20%			
Median	-0.25%	0.30%	0.55%	-0.80%	0.81%	1.62%	1.07%	
Percent above median	47.17%	52.83%	5.66%	43.54%	57.02%	13.48%	7.82%	
Net about affallages								
Net charge-offs/loans Mean	0.39%	0.18%	-0.21%	0.74%	0.21%	-0.54%	-0.32%	-6.20
		0.18%	-0.21/6	0.74%	0.21%	-0.54 /6	-0.32 /6	-0.20
Standard Error	0.02%	0.02%		0.04%	0.02%			
Median	0.03%	-0.02%	-0.05%	0.07%	-0.06%	-0.13%	-0.07%	
Percent above median	53.51%	46.32%	-7.19%	55.30%	44.45%	-10.85%	-3.66%	
1 1								
Loan loss provisions/io		0.17%	-0.20%	0.81%	0.23%	-0.58%	-0.38%	-7.15
Mean	0.37%		-0.20%		0.23%	-0.50%	-0.36%	-7.15
Standard Error	0.02%	0.02%		0.04%	0.02%			
Median	0.02%	-0.02%	-0.04%	0.06%	-0.04%	-0.10%	-0.06%	
Percent above median	52.59%	47.23%	-5.36%	54.26%	45.49%	-8.78%		
. Crook was a modern	52.55 /6	.,.23,0	2.23,0	÷= ÷ /•				

Notes: Growers are banking companies with a growth rate in total assets above the median in that state and year; shrinkers are banking companies with a growth rate in total assets below the median in that state and year (the growth of assets at banks which exit the industry is coded as -100 percent). All of the variables are measured relative to the median value of that variable in the applicable state and year.

Table 7

Gressions of Banking Company Growth on Ex-Ante Performance indicators

Regressions of Banking Company Gro	wth on Ex-Ante F	<u>Performance indic</u>		
			Pooled	
	_		with Bank	
	Pre-	Post-	Fixed	
	Deregulation	Deregulation	Effects	
	(1)	(2)	(3)	
Specification based on Return on Assets				
Above Median Return on Assets Indicator	-0.0262 *	-0.0056	0.0148	
	(0.0081)	(0.0108)	(0.0099)	
Above Median Return on Assets Indicator	•	•	`0.0606 ^{´ ±}	
* Post Deregulation Indicator		•	(0.0146)	
Capital-Asset Ratio	2.7751 *	2.1414 *	0.2212	
Capital-Asset Hatio	(0.1725)	(0.2067)	(0.2404)	
Log of Total Accete	0.0069	-0.0012	-0.8491 *	
Log of Total Assets			(0.0241)	
••	(0.0040)	(0.0060)	** * * * * * * * * * * * * * * * * * * *	
N	6,968	5,888	4,521	
R-Squared	22.13%	17.13%	44.80%	
Specification based on Return on Equity				
Above Median Return on Equity Indicator	0.0087	0.0326 *	0.0177	
renewal transfer of the second	(0.0078)	(0.0103)	(0.0102)	•
Above Median Return on Equity Indicator	(0.007.0)	(0.0100)	0.0650 *	
	_	_	(0.0148)	
* Post Deregulation Indicator	0.7000 *	0.1001 *	0.3028	
Capital-Asset Ratio	2.7338 *	2.1381 *		
	(0.1715)	(0.2017)	(0.2384)	
Log of Total Assets	0.0071	-0.0040	-0.8556 *	
	(0.0041)	· (0.0060)	(0.0242)	
N	6,968	5,888	4,521	
R-Squared	22.02%	17.27%	44.98%	
Specification based on Net Charge-Offs/Loans				
Above Median Charge-offs Indicator	-0.0428 *	-0.0758 *	-0.0012	
ADOVE MEDIAN Charge-one moleculor	(0.0086)	(0.0097)	(0.0106)	
About Marillan Ohanna offa Indiantes	(0.0000)	(0.0097)	-0.0773 *	
Above Median Charge-offs Indicator	-	•		
* Post Deregulation Indicator		4.0700	(0.0141)	
Capital-Asset Ratio	2.4162 *	1.6788 *	0.0803	
	(0.1879)	(0.2205)	(0.2677)	
Log of Total Assets	0.0104 *	-0.0089	-0.8963 *	
	(0.0047)	(0.0062)	(0.0247)	
N	5,990	5,755	3,992	
R-Squared	18.14%	16.17%	47.32%	
Specification based on Loan Loss Provisions/Loans				
Above Median Loan Loss Provisions Indicator	-0.0055	-0.0359 *	0.0042	
ADDAG MAGNIGH FORM FORM LIAMSTONS INDICATOR	(0.0078)	(0.0098)	(0.0098)	
About Madian Loop Long Descriptions Indicates	(0.0070)	(0.0000)	-0.0584 *	
Above Median Loan Loss Provisions Indicator	-			
* Post Deregulation Indicator		4 2000 *	(0.0140)	
Capital-Asset Ratio	2.6940 *	1.8838 *	0.1721	
	(0.1724)	(0.2221)	(0.2569)	
Log of Total Assets	0.00702	-0.00442	-0.8397 *	
	(0.0040)	(0.0058)	(0.0240)	
N	6,977	5,925	4,554	
R-Squared	21.96%	15.78%	44.28%	
· ·				

Notes: Columns (1) and (2) present regressions for the pre and post branching deregulation periods, respectively; each regression includes a constant (not shown). Column (3) presents the results of the pooled model estimated with fixed effects; this model also includes a constant and post-deregulation indicator (not shown). The number of observations in this model represents the number of banking companies that appear in both samples, and the R-Squared represents the fit of the "within" estimator. Each set of rows represents the results using one proxy for bank quality. The performance indicators are measured as of the beginning of the period; the indicator equals 1 for banking companies with above-median values for that variable in the appropriate state and year. Since states deregulated at different times, each cross-sectional regression includes a set of state indicator variables (not shown). Also, we drop banking companies with capital-asset ratios below 3 percent from the two profit regressions. White standard errors are reported below each coefficient in parentheses. Coefficients denoted with a "*" are statistically significant at the 5 percent level.

FEDERAL RESERVE BANK OF NEW YORK RESEARCH PAPERS 1996

The following papers were written by economists at the Federal Reserve Bank of New York either alone or in collaboration with outside economists. Single copies of up to six papers are available upon request from the Public Information Department, Federal Reserve Bank of New York, 33 Liberty Street, New York, NY 10045-0001 (212) 720-6134.

- 9601. Bartolini, Leonardo, and Gordon M. Bodnar. "Are Exchange Rates Excessively Volatile? And What Does 'Excessively Volatile' Mean, Anyway?" January 1996.
- 9602. Lopez, Jose A. "Exchange Rate Cointegration Across Central Bank Regime Shifts." January 1996.
- 9603. Wenninger, John, and Daniel Orlow. "Consumer Payments Over Open Computer Networks." March 1996.
- 9604. Groshen, Erica L. "American Employer Salary Surveys and Labor Economics Research: Issues and Contributions." March 1996.
- 9605. Uctum, Merih. "European Integration and Asymmetry in the EMS." April 1996.
- 9606. de Kock, Gabriel S. P., and Tanya E. Ghaleb. "Has the Cost of Fighting Inflation Fallen?" April 1996.
- 9607. de Kock, Gabriel S. P., and Tania Nadal-Vicens. "Capacity Utilization-Inflation Linkages: A Cross-Country Analysis." April 1996.
- 9608. Cantor, Richard, and Frank Packer. "Determinants and Impacts of Sovereign Credit Ratings." April 1996.
- 9609. Estrella, Arturo, and Frederic S. Mishkin. "Predicting U.S. Recessions: Financial Variables as Leading Indicators." May 1996.
- 9610. Antzoulatos, Angelos A. "Capital Flows and Current Account Deficits in the 1990s: Why Did Latin American and East Asian Countries Respond Differently?" May 1996.

- 9611. Locke, Peter R., Asani Sarkar, and Lifan Wu. "Did the Good Guys Lose? Heterogeneous Traders and Regulatory Restrictions on Dual Trading." May 1996.
- 9612. Locke, Peter R., and Asani Sarkar. "Volatility and Liquidity in Futures Markets." May 1996.
- 9613. Gong, Frank F., and Eli M. Remolona. "Two Factors Along the Yield Curve." June 1996.
- 9614. Harris, Ethan S., and Clara Vega. "What Do Chain Store Sales Tell Us About Consumer Spending?" June 1996.
- 9615. Uctum, Merih, and Michael Wickens. "Debt and Deficit Ceilings, and Sustainability of Fiscal Policies: An Intertemporal Analysis." June 1996.
- 9616. Uctum, Merih, and Michael Aglietta. "Europe and the Maastricht Challenge." June 1996.
- 9617. Laster, David, Paul Bennett, and In Sun Geoum. "Rational Bias in Macroeconomic Forecasts." July 1996.
- 9618. Mahoney, James M., Chamu Sundaramurthy, and Joseph T. Mahoney. "The Effects of Corporate Antitakeover Provisions on Long-Term Investment: Empirical Evidence." July 1996.
- 9619. Gong, Frank F., and Eli M. Remolona. "A Three-Factor Econometric Model of the U.S. Term Structure." July 1996.
- 9620. Nolle, Daniel E., and Rama Seth. "Do Banks Follow Their Customers Abroad?" July 1996.
- 9621. McCarthy, Jonathan, and Charles Steindel. "The Relative Importance of National and Regional Factors in the New York Metropolitan Economy." July 1996.
- 9622. Peristiani, S., P. Bennett, G. Monsen, R. Peach, and J. Raiff. "Effects of Household Creditworthiness on Mortgage Refinancings." August 1996.
- 9623. Peristiani, Stavros. "Do Mergers Improve the X-Efficiency and Scale Efficiency of U.S. Banks? Evidence from the 1980s." August 1996.
- 9624. Ludvigson, Sydney. "Consumption and Credit: A Model of Time-Varying Liquidity Constraints." August 1996.

- 19625. Ludvigson, Sydney. "The Channel of Monetary Transmission to Demand: Evidence from the Market for Automobile Credit." August 1996.
 - 9626. Sobol, Dorothy M. "Central and Eastern Europe: Financial Markets and Private Capital Flows." August 1996.
 - 9627. Evans, Joan, and James M. Mahoney. "The Effects of Daily Price Limits on Cotton Futures and Options Trading." August 1996.
 - 9628. Molyneux, Philip, and Rama Seth. "Foreign Banks, Profits and Commercial Credit Extension in the United States." August 1996.
 - 9629. Cantor, Richard, and Robert Driskill. "Can a Fiscal Contraction Strengthen a Currency? Some Doubts About Conventional Mundell-Fleming Results." August 1996.

To obtain more information about the Bank's RESEARCH PAPERS series and other publications, visit the Federal Reserve Bank of New York's site on the World Wide Web (http://www.ny.frb.org). From the research publications page, you can view abstracts of RESEARCH PAPERS and STAFF REPORTS and order the full-length, hard-copy versions of them electronically. Interested readers can also view and download articles from CURRENT ISSUES IN ECONOMICS AND FINANCE and the ECONOMIC POLICY REVIEW.